Author’s response to reviews

Title: Transcriptome sequencing of lncRNA, miRNA, mRNA and interaction network constructing in Coronary Heart disease

Authors:

Jiangquan Liao (liaojiangquan@163.com)
Jie Wang (wangjie0103@126.com)
Yongmei Liu (lymsd@163.com)
Jun Li (doctorli1031@163.com)
Lian Duan (popale2003@163.com)

Version: 3 Date: 08 Aug 2019

Author’s response to reviews:

To David Tan (Reviewer 1):

We have read and discussed your comments again and again. We truly appreciate your work in reviewing our manuscript. And your comments have brought many thinking and suggestions to our research group. We would definitely improve our study in the future. It is a pity that we could not fully address all of your concerns due to some limitation. Thank you for your understanding and dedication.

To editor:

Thank you for the opportunity for our work to be published in BMC medical genomics. We have made all the revisions you have required. Please let us know if there is anything we should do.

And if any further contact should be made, would you please send a copy to Jiangquan Liao (liaojiangquan@163.com)? Since he is responsible for the revision and the publication fee transfer.

Yours sincerely,

Jiangquan Liao et al